Author’s response to reviews

Title: Genetic Determinants of Sporadic Breast Cancer in Sri Lankan Women

Authors:

Nirmala Sirisena (nirmalasirisena@yahoo.com)
Adebowale Adeyemo (adeyemoa@mail.nih.gov)
Anchala Kuruppu (anchalak@gmail.com)
Nilaksha Neththikumara (nilakshafreezon@gmail.com)
Nilakshi Samaranayake (nilakshis35@gmail.com)
Vajira Dissanayake (vajirahwd@hotmail.com)

Version: 3 Date: 05 Feb 2018

Author’s response to reviews:

5th February 2018

Editor
BMC Cancer

Dear Sir

Response to Editor’s comments on the manuscript titled “Genetic Determinants of Sporadic Breast Cancer in Sri Lankan Women” – BCAN-D-17-00449R3

Thank you for giving us the opportunity to submit a revised version of the manuscript. A detailed description is annexed to this letter explaining how we have dealt with each of the editor’s concerns. The revised manuscript is also attached.

We look forward to an early response.

Thank you.

Sincerely,

Dr. Nirmala Sirisena
Editor Comments:

1. We note that the current submission contains some textual overlap, in the discussion section particularly, with other previously published works. Please refer to the attached iThenticate document and re-phrase these sections in your own words to ensure no overlap. If there is overlap in the Methods section, please ensure that you summarize the methods and cite the source.

The textual overlaps in the discussion section and other relevant sections have been re-phrased in the revised manuscript.